Author's response to reviews

Title: 23 Gauge Pars Plana Vitrectomy For The Removal Of Retained Intraocular Foreign Bodies

Authors:

Kemal Yuksel (drkemal68@gmail.com)
Ugur Celik (h.ugurcelik@gmail.com)
Cengiz Alagoz (alagozcengiz@gmail.com)
Okkes Baz (dr.bazokkes@hotmail.com)
Burcu Celik (n.burcuguleryuz@gmail.com)
Ahmet Taylan Yazici (ahmettaylan19@gmail.com)

Version: 6
Date: 29 May 2015

Author's response to reviews: see over
Dear Editor

I have done the necessary corrections on the paper; here you can see all the comments that I have made.

1. Title: ?Ppv? should be ?pars plana vitrectomy? 

   The mistake was corrected.

2. Reference 2, 7, 21, 23, 24, 30, 32, 41: The style is still different from others. Be careful. For example, 2008 Feb;246(2):181-6 should be 2008;246:181?186.

   The mistakes were corrected.

3. Table 4: 20/200 should be >20/200

   Dear esteemed reviewer we agree with you about our mistake and decided to remove the Table 4, and renamed the other tables.

4. Table 4: What is the meaning of P<0.001? You showed that in 12 out of the 14 eyes whose preoperative VA was <20/200, the postoperative VA was still <20/200. I assume there was no statistical difference in VA distribution before and after surgery.

   Dear esteemed reviewer we agree with you about our mistake and decided to remove the Table 4, and renamed the other tables.

5. Table 5: >0.05 should be <0.05.

   The mistake was corrected.

6. Requesting for Copy-Edit

   A native speaker reviewed the paper.

Kindest Regards.

Ugur Celik